

Decomposing the Effects of Digitalization on Workers' Job Satisfaction

Working Paper

Author(s): Bolli, Thomas (b; Pusterla, Filippo (b)

Publication date: 2021-03

Permanent link: https://doi.org/10.3929/ethz-b-000473930

Rights / license: In Copyright - Non-Commercial Use Permitted

Originally published in: CES Working Papers 4

This page was generated automatically upon download from the <u>ETH Zurich Research Collection</u>. For more information, please consult the <u>Terms of use</u>.



CES Working Paper Series

Decomposing the Effects of Digitalization on Workers' Job Satisfaction

Thomas Bolli & Filippo Pusterla CES Working Paper, No. 4, 2021



Decomposing the Effects of Digitalization on Workers' Job Satisfaction^{*}

Thomas Bolli & Filippo Pusterla[‡]

This version: March 10, 2021

Abstract

This paper provides a conceptual framework for analyzing the importance of the multiple channels through which digitalization affects job satisfaction. Our framework investigates four distinct groups of channels allowing digitalization to shape job satisfaction: change in time use, creation of new activities, access to information, and adoption of communication tools. Using graduates of professional education and training colleges in Switzerland as a case study, we investigate the relative strength of the channels through which digitalization affects job satisfaction. We find that digitalization increases job satisfaction mainly through the creation of new activities, specifically by increasing productivity and making work more interesting. Our results further suggest that among the channels negatively affecting job satisfaction, increase of time pressure and worsening of work-life balance are much more important than the threat of losing one's job. Furthermore, we present evidence on the heterogeneity of these results across gender, age, management position, and field of study.

JEL-Classification: J28, O33

Keywords: digitalization, job satisfaction.

^{*}Acknowledgments: We thank ODEC, Umbrella organization of graduates from professional education and training (PET) colleges in Switzerland, for the opportunity to pose specific questions on the subject of digital transformation in the workplace as experienced by PET graduates in the context of the ODEC Salary Survey 2019. We are grateful for comments from Ursula Renold, Reto Odermatt, Thomas Dohmen, and participants at the CES-Leading House Economics of Education joint Workshop.

[‡]ETH Zurich, Chair of Education System, Leonhardstrasse 21, 8092 Zurich, Switzerland. Corresponding address: filippo.pusterla@mtec.ethz.ch

1 Introduction

Digitalization is the rapidly growing sociotechnical phenomenon of adopting information and communication technologies (ICT) (Legner et al., 2017). Most of the economic literature analyzing the labor market effects of digitalization focuses on the number of jobs that new technologies replace (e.g., Autor, 2015; Frey & Osborne, 2017; Acemoglu & Restrepo, 2018; Graetz & Michaels, 2018). Yet relatively little attention has been paid to the effects of ICT adoption on jobs not replaced by digitalization, with only limited evidence on the mechanisms through which digitalization affects workers' job satisfaction.

However, firms' ability to assess the way in which digitalization affects job satisfaction is crucial, because understanding through which channels digitalization affects workers' job satisfaction might help them better evaluate the introduction of new technologies. Likewise, workers' knowing how digitalization will affect their job satisfaction might help them to assess the consequences of increasingly diffuse work practices (e.g., home offices).

Theoretically, digitalization can affect workers' job satisfaction either positively (e.g., by decreasing the percentage of repetitive tasks and increasing that of interesting ones) or negatively (e.g., by increasing the level of stress or decreasing work-life balance). A growing body of literature at the intersection of economics and psychology suggests an overall positive effect of digitalization on workers' job satisfaction and well-being (e.g., McMurtrey et al., 2002; Salanova et al., 2004; Golden & Veiga, 2005; Day et al., 2010; Limbu et al., 2014). However, no study looking at how digitalization might affect job satisfaction has yet examined more than one channel through which that effect might operate. For example, Moqbel et al. (2013) highlight the role of social networks in increasing workers' job satisfaction, while Martin & Omrani (2015) show that information technology use positively affects job satisfaction due to an increase in labor productivity. Thus far, no paper systematically identifies and assesses the multiple channels through which digitalization affects job satisfaction.

This paper provides a comprehensive framework of the channels through which digitalization may affect job satisfaction. In all channels, digitalization affects job satisfaction by first changing some characteristic of the job itself, and then that change impacts the worker's satisfaction. Therefore, all of the channels through which digitalization might affect job satisfaction are changes in job characteristics caused by digitalization. Castellacci & Tveito (2018) argue that digitalization shapes workers' job characteristics in four main ways: change in time use, creation of new activities, access to information, and use of communication tools. We further subdivide these broad dimensions into 10 specific ways that digitalization affects job characteristics. Based on the existing evidence, we formulate hypotheses on how the 10 channels affect job satisfaction.

Specifically, we hypothesize that digitalization decreases job satisfaction by increasing time pressure at work, by increasing the fear of losing one's job, by deteriorating work-life balance, and by smoothing the transition between working hours and leisure time. Conversely, we hypothesize that digitalization increases job satisfaction by making work more interesting, by reducing the proportion of repetitive tasks, by increasing productivity, and by increasing autonomy at work. Furthermore, we hypothesize that digitalization also increases job satisfaction by making forms of working more flexible and by simplifying interactions with colleagues and superiors.

We empirically test our hypotheses by using a survey conducted among students and graduates of professional education and training (PET) colleges in Switzerland in 2019. Beyond general information on workers, our survey contains specific questions on digital transformation in the workplace, including asking respondents to evaluate statements about the effects of digitalization on different job characteristics and to self-assess the effect of digitalization on their job satisfaction. Having information on both the total effect of digitalization on job satisfaction and the effect of digitalization on single job characteristics allows us to assess the relative importance of the channels through which digitalization affects job satisfaction.

Our results suggest that digitalization increases job satisfaction among PET graduates by increasing work productivity, making work more interesting, fostering interactions with coworkers and supervisors, increasing workers' autonomy, and allowing flexible forms of work. Furthermore, our results suggest only a moderate negative effect of digitalization on job satisfaction through an increase in time pressure. However, we find that digitalization negatively affects job satisfaction by worsening work-life balance but not by smoothing the transition between working hours and leisure time. Finally, our estimates provide no evidence that digitalization positively affects job satisfaction by reducing repetitive tasks. Although the widespread notions that the fear of losing one's job to digitalization negatively affects job satisfaction is confirmed, it remains small in magnitude in our sample.

Furthermore, heterogeneity analyses on subset of workers suggest that the worsening of the work-life balance is more relevant for men, for workers aged more than 35 years (roughly the average age in our sample), for workers with an executive position, and for workers whose field of study is technology-related. For the interestingness of work, we find a larger effect for males and for workers er than 35. In contrast, the effect that digitalization has on job satisfaction through an increase in autonomy is lower for women, for young workers, and for workers who did not study in technology-related fields. In terms of productivity, we find that digitalization is more beneficial for women, for older workers, for workers without an executive position, and for workers who did not study in technology-related fields. Finally, the positive effect that digitalization has on job satisfaction by simplifying interactions with colleagues and superiors is larger for non-executive workers than for executives.

This paper contributes both theoretically and empirically to the current debate on the impact of digitalization on job satisfaction. We first build a comprehensive framework of the channels identified in the literature and formulate hypotheses for each of these channels. Second, we provide empirical evidence on the relative importance that these channels have in explaining the effect of digitalization on workers' job satisfaction.

The rest of the paper is organized as follows. Section 2 presents the conceptual framework of the study and derives the hypotheses. Section 3 explains the estimation strategy, and Section 4 describes the data set. Section 5 presents the results and discusses the heterogeneity across workers. Section 6 concludes and discusses implications for future research.

2 Theoretical Framework and Hypotheses

A growing body of literature at the intersection of economics and psychology suggests a positive relationship between digitalization and workers' job satisfaction (e.g., McMurtrey et al., 2002; Salanova et al., 2004; Golden & Veiga, 2005; Day et al., 2010; Limbu et al., 2014; Martin & Omrani, 2015). However, no study in this literature has yet analyzed the channels through which digitalization affects job satisfaction in a comprehensive framework. To fill this gap, this paper decomposes the effects of digitalization on workers' job satisfaction into different channels and assesses their importance relative to one another.

To identify the channels through which digitalization affects job satisfaction, we use Castellacci & Tveito's (2018) theoretical model, which groups these channels into four distinct dimensions, which Figure 1 shows. First, while digitalization increases efficiency and frees up time, it can make some occupations obsolete. Thus, digitalization has an effect on job satisfaction through the "change in time use" dimension. Second, digitalization can create new activities that provide both security and personal control, in turn leading to a positive effect on job sat-





Source: Authors' depiction based on Castellacci & Tveito (2018)

isfaction and well-being. Digitalization has thus an effect on job satisfaction through the "new activities" dimension.

Third, digitalization enables individuals to obtain, access, process, and archive information much more systematically and rapidly than previously possible. Easier information access improves quality of work and eventually workers' job satisfaction. Digitalization thus has an effect on job satisfaction through the "access of information" dimension. Fourth, while digitalization increases the opportunities for communication and eventually fosters social capital and knowledge sharing, it also distracts workers and reduces their efficiency. Digitalization thus has an effect on job satisfaction through the "communication tools" dimension.

To understand the mechanisms through which digitalization affect workers' job satisfaction, we subdivide Castellacci & Tveito's (2018) four-dimensional theoretical model into 10 specific channels. Each of these describes one job characteristics affected by digitalization, and the changes in job characteristics drive changes in workers' job satisfaction. Specifically, we consider the following 10 job characteristics affected by digitalization as channels through which digitalization affects job satisfaction: Time pressure, fear of losing one's job, work-life balance, smoothness of transition between work and private life, interestingness of tasks, productivity, autonomy, working time flexibility, and the simplicity of interaction with colleagues and superiors. For simplicity, hereafter we refer to the effect that digitalization has on job satisfaction

Figure 2: Summary of hypotheses



through a job characteristic as "channels". A channel describes thus digitalization's effect on job satisfaction via a change in a specific job characteristic.

Figure 2 previews the hypotheses that we propose in this paper. The 10 job characteristics are grouped according to the four dimensions developed by Castellacci & Tveito (2018). The left side shows the hypothesized effect of digitalization on the 10 job characteristics (e.g., the effect of digitalization on time pressure at work). The right side displays the hypothesized effect of the 10 job characteristics on job satisfaction (e.g. the effect of time pressure at work on job satisfaction). The combination of these effects yields the effect of digitalization on job satisfaction for each job characteristic and, therefore, each channel.

By formulating our hypotheses, we refer on the entire sample and refrain from refining them according to workers' individual characteristics. Nevertheless, in Section 5.2 we present and discuss the heterogeneity of our results according to gender, age, management position, and field of study.

Change in Time Use

The dimension change in time use includes four channels (in our model, 1-4): time pressure, fear of job loss, work-life balance, and transition smoothness between work and private life. The

first channel, time pressure, captures the possibility that digital technologies at work can expose employees to working under pressure, having frequent tight deadlines resulting from electronic workflows, and lacking sufficient time for carrying out daily tasks (Agypt & Rubin, 2012). These conditions create "technostress", which Tarafdar et al. (2010) define as the psychological effects stemming from the inability to cope with computer or software use at work. A large literature shows that technostress negatively affects job satisfaction (e.g., Tarafdar et al., 2007; Ragu-Nathan et al., 2008; Ayyagari et al., 2011). We therefore hypothesize as follows:

H1: Digitalization decreases job satisfaction by increasing time pressure at work.

The second channel, fear of job loss, captures the likelihood that digitalization will make certain jobs obsolete (Rotman, 2013; Autor, 2014 & 2015). Research shows that the perception of job insecurity is an important factor in stress (Hartley et al., 1990), which is negatively related to job satisfaction (Reisel et al., 2010). We therefore hypothesize as follows:

H2: Digitalization decreases job satisfaction by increasing the fear of losing one's job.

The third channel, work-life balance, captures the deterioration of work-life balance created by digitalization (Nam, 2014). The literature suggests that a worse work-life balance negatively affects subjective job satisfaction (Gallie & Russell, 2009; Scandura & Lankau, 1997). We therefore hypothesize as follows:

H3: Digitalization decreases job satisfaction by deteriorating the work-life balance.

The fourth channel, transition smoothness between work and private life, captures digitalization's allowing a smoother transition between working hours and leisure time (Boswell & Olson-Buchanan, 2007). A smoother transition between the two can have negative consequences for job satisfaction, for example, by exacerbating the work-family conflict (Boswell & Olson-Buchanan, 2007). We therefore hypothesize as follows:

H4: Digitalization decreases job satisfaction by smoothing the transition between working hours and leisure time.

New Activities

The dimension new activities includes four channels (in our model, 5-8): interestingness of tasks, percentage of repetitive tasks, productivity, and autonomy. The first of these channels (i.e., the

fifth channel), interestingness of tasks, captures the way in which digitalization leads to the creation and development of new working activities and tasks (Carlsson, 2004). These new activities often require specific skills, provide physical security, and increase personal control, all factors that are positive for job satisfaction (Warr, 2003; Castellacci & Viñas-Bardolet, 2019). We therefore hypothesize as follows:

H5: Digitalization increases job satisfaction by making work more interesting.

The sixth channel, percentage of repetitive tasks, captures the effect of digitalization in reducing the proportion of repetitive tasks and physically straining labor (Acemoglu & Autor, 2011). Such a reduction allows workers to allocate more time to more rewarding activities (Askenazy & Caroli, 2010), an outcome that, in turn, has a positive effect on workers' job satisfaction (Melamed et al., 1995; Kristensen & Johansson, 2008). Nevertheless, the introduction of new tasks might also increase the level of job stress, in turn negatively influencing job satisfaction (Konradt et al., 2003; Morris & Venkatesh, 2010). However, Castellacci & Viñas-Bardolet (2019) suggest that the effect of a reduction in repetitive tasks on job satisfaction is particularly positive for white-collar workers. As the survey sample in this paper consists of tertiary-educated workers, we favor the argument of more rewarding activities. We thus formulate the following hypothesis:

H6: Digitalization increases job satisfaction by reducing the proportion of repetitive tasks.

The seventh channel, productivity, shows that digitalization allows more productive activities (Brynjolfsson & McAfee, 2011). Activities that are more productive imply higher wages (all else being equal), which in turn lead to higher job satisfaction (D'Addio et al., 2007; Castellacci & Viñas-Bardolet, 2019). We therefore hypothesize as follows:

H7: Digitalization increases job satisfaction by increasing productivity.

The eighth channel, autonomy, captures the effect of digitalization on employees' autonomy at work. Mazmanian (2013) provides evidence that digital devices do not limit workers' discretion, freedom, or authority but instead enhance their autonomy. Bloom et al. (2014) further suggest that ICT makes accessing information less expensive, thereby giving workers more autonomy and a wider span of control. ICT thus acts as a decentralizing force that allows workers to handle situations more autonomously. Furthermore, the literature shows that workers with a higher degree of autonomy are typically more satisfied (Golden & Veiga, 2005; Lopes et al., 2014). We therefore hypothesize as follows:

H8: Digitalization increases job satisfaction by increasing autonomy at work.

Access to Information

The dimension access to information has only one channel (in our model, 9). This ninth channel, working time flexibility, captures the way digitalization improves employees' access to information, with an increasing number of tasks no longer requiring a specific workstation (Popma, 2013). Workplace-independent access to information enables more flexible working time (Duxbury et al., 2007), and Raziq & Maulabakhsh (2015) show that flexible working hours increase job satisfaction. Similarly, Kelliher & Anderson (2010) find that flexible workers report higher levels of job satisfaction than their non-flexible counterparts. We therefore hypothesize as follows:

H9: Digitalization increases job satisfaction by making forms of working more flexible.

Communication Tools

The dimension communication tools has only one channel (in our model, 10). This tenth channel, simplicity of interaction with colleagues and superiors, captures digitalization's simplifying the interactions between individuals. For example, Koku et al. (2001) highlight the positive effect of the Internet in facilitating and maintaining off-line relationships. Zhao (2006) finds that individuals using the Internet for interpersonal contact usually have more social ties than those who do not. Furthermore, digital technologies also simplify workplace interaction. Moqbel et al. (2013) focus on the role of social networking sites (SNS), which are web-based services that allow workers to build social networks or relationships with other people. They find that the use of SNS at work increases organizational commitment and job satisfaction.

In a sample covering 13 countries, Amichai-Hamburger & Hayat (2011) investigate the influence of Internet use on social interactions, finding that Internet usage is positively correlated with the socially related interactions of people in the same profession. In turn, simplified interactions with colleagues and superiors increase workers' job satisfaction (Pincus, 1986; Warr, 2003). Additionally, Intranet use at work has been found to positively affect the sharing of internal knowledge within a firm (Hendriks, 1999), and knowledge sharing improves social capital (Huysman & Wulf, 2006) and increases work quality (Haas & Hansen, 2007), in turn increasing job satisfaction (Requena, 2003).

Nevertheless, some recent studies also show that the use of communication tools at work (e.g., Facebook) can have negative effects on productivity, in turn negatively affecting workers' morale and job satisfaction (Brooks, 2015). However, despite these new contradictory results, we favor the argument of enhanced communication because more largely documented by the literature. We therefore hypothesize as follows:

H10: Digitalization increases job satisfaction by simplifying worker interactions with colleagues and superiors.

3 Empirical strategy

This section presents the empirical strategy we use to assess the relative importance of the channels through which digitalization affects job satisfaction. To do so, we start by presenting a structural model and discussing the challenges that such an approach could pose. We then apply a reduced form model, which allows us to test our hypotheses and poses fewer challenges to both measuring digitalization intensity and assessing job satisfaction.

3.1 Structural Model

Identifying the influence of each channel in a structural model requires estimating multiple equations. First, we need to estimate the effect of digitalization on 10 job characteristics for worker i as represented by the following system of equations:

$$Job \ Characteristic_i^c = \phi_c + \theta_c \ Digitalization_i + \vartheta_c \ X_i + \tau_{ci} \tag{1}$$

where $Digitalization_i$ stands for the digitalization of worker *i*'s job, X_i is a vector of other variables that affect job satisfaction of worker *i*, and τ is the error term.

Second, the structural model contains an estimation of the relationship between the 10 job characteristics c and the job satisfaction of worker i:

$$Job \ Satisfaction_i = \alpha + \sum_{c=1}^{10} \beta_c Job \ Characteristic_i^c + \eta \ X_i + \vartheta_i$$
(2)

where β_c reflects the impact of job characteristic c on job satisfaction. The vector X is defined as above, while ϑ is the error term.

However, estimating this structural model faces a number of challenges in terms of measuring the variables in equations 1 and 2. An empirical challenge involves the difficulty in measuring digitalization. The literature often measures digitalization by counting the number of computer or digital devices (Caselli & Coleman, 2001). Nevertheless, the stock of computers measures digitalization imperfectly, because it measures only the availability of computers, not their effective use by workers. Therefore, we apply a reduced form model, which allows us to identify the relative importance of each channel.

3.2 Reduced Form Model

Inserting equation 1 into equation 2 and taking the first derivative with respect to digitalization yields the following reduced form:

$$\omega_i = \frac{\partial Job \ Satisfaction_i}{\partial Digitalization_i} = \sum_{c=1}^{10} \beta_c \theta_{ci} + \epsilon_i \tag{3}$$

where ω_i is the partial derivative of job satisfaction with respect to digitalization. θ_c denotes the effect of digitalization on job characteristic c. β_c reflects the impact of job characteristic c on job satisfaction.

We operationalize ω_i by asking respondents how strongly digitalization affects his or her job satisfaction, measured on a five-point Likert scale. Similarly, we operationalize θ_c by asking respondents to assess the impact of digitalization on the corresponding job characteristic c. We thus estimate via OLS the following equation:

$$\widetilde{\omega}_i = \sum_{c=1}^{10} \beta_c \widetilde{\theta_{ci}} + \gamma \widetilde{X}_i + \epsilon_i \tag{4}$$

where the superscript ~ describes parameters that have been self-assessed by respondents. This equation also account for other workers' characteristics that might affect job satisfaction but are unrelated to digitalization. Specifically, \widetilde{X}_i is a vector of control variables, a vector with the following worker characteristics: $age, age^2, gender$, a dummy for an executive position, 8 dummies for the field of study, and 13 dummies for the industry. Finally, the identification of β assumes that job characteristics c are orthogonal to any other potential characteristics through which digitalization affects job satisfaction. To account for this potential source of omitted variable bias, \widetilde{X}_i further includes a variable that captures how strongly respondents assess the impact of digitalization on their job in the previous year, measured on a five-point Likert scale. ϵ is the error term that is estimated robust.

Estimating equation 4 via OLS yields estimates for the impact of job characteristic c on job satisfaction. To analyze the effect of digitalization on job satisfaction, we multiply for each worker characteristic c the estimated $\hat{\beta}_c$ with the corresponding $\tilde{\theta}_c$. While the calculation of $\hat{\beta}_c \tilde{\theta}_c$ is straightforward, its interpretation is far from trivial. Indeed, this measure combines the effect of job characteristic c on job satisfaction ($\hat{\beta}_c$) and the extent to which workers agree with their survey assessment of the impact of digitalization on this job characteristic ($\tilde{\theta}_c$). Therefore, to simplify the interpretation, we decompose the overall goodness of fit R^2 into the explanatory power of individual regressors. The decomposition of R^2 translates into the importance of the different regressors by giving a measure that is more easily interpreted.

One convenient measure for decomposing the overall goodness of fit is the Shapley value (Shapley, 1953), which computes the contribution of a single variable to the goodness-of-fit of a statistical model. Assume, for example, a full regression model with k explanatory variables $(x_1, x_2, ..., x_k)$. According to Huettner et al. (2012), to calculate the contribution of each variable, we need to estimate all possible submodels derived by the permutation of the regressors. Mathematically, to calculate the contribution of a given regressor j we need to estimate the same number of submodels as the number of permutations (K!) of k regressors:

$$R_j^2 = \frac{1}{K!} R^2(f(x_j^{\mu}, x_j)) - R^2(f(x_j^{\mu}))$$
(5)

where μ maps all K! variable permutations. By subtracting the R^2 of the model not including x_j from the R^2 of the model including x_j and all regressors preceding x_j in that particular order (x_j^{μ}) , we obtain the Shapley value, which measures j's average marginal contribution to R^2 across all possible permutations.

4 Data and description of variables

The data stems from the ODEC Salary Survey conducted as an online survey among students and graduates of Swiss professional education and training (PET) colleges in 2019. This formal vocational tertiary education at level 6 of the ISCED-2011 classification takes from two to four years, depending on the PET college and on whether the education is full-time or part-time. While students account for about 10% of the sample and have a response rate of about 20%, graduates account for the remaining 90%, with a response rate of about $11\%^{1}$.

Table 1 shows the summary statistics of variables used in the estimation. The dependent variable measures the influence of digitalization on job satisfaction on a five-point Likert scale (1="less satisfied"; 3="no change"; 5="more satisfied"). The mean of 3.47 suggests that digitalization on average increases job satisfaction of workers with a PET college diploma. This persistent positive effect of digitalization on workers' job satisfaction—in line with the findings in literature (e.g., McMurtrey et al., 2002; Salanova et al., 2004; Golden & Veiga, 2005; Day et al., 2010; Limbu et al., 2014; Martin & Omrani, 2015)—needs cautious interpretation, because it is specific to the subsample in this paper.

Breakdowns of the dependent variable by gender, age, management position, and field of study yield values above 3, suggesting an overall positive effect of digitalization on job satisfaction. Nevertheless, some differences are noteworthy. Men report a larger positive effect of digitalization on job satisfaction than women. Additionally, workers younger than age 35 also report higher levels of job satisfaction than do older workers. We also observe small differences between workers in executive positions, who report a slightly larger positive effect than nonexecutive workers. Finally, across fields of study, we find that workers in the fields of social work and adult education report almost no change in digitalization-induced job satisfaction, whereas workers from the fields of arts and business administration report a relatively large positive effect.

The main explanatory variables capture, on a five-point Likert scale, to what extent respondents agree with statements about the impact of digitalization on the 10 job channels through which we hypothesize that digitalization affects job satisfaction (1="I don't agree at all"; 5="Ifully agree"). The results suggest that the strongest effect of digitalization lies in increasingproductivity (3.65), followed by simplifying interactions with colleagues and superiors (3.41)and making work more interesting (3.4). Moreover, we find an average effect in terms of anincrease in more flexible forms of working time (3.25), a reduction in the proportion of repetitivetasks (3.24), an increase in time pressure (3.23), an increase in autonomy (3.15), and a smoothtransition between working hours and leisure time (3.01). The least strong effects appear interms of worsening work-life balance (2.72) and fear of losing one's job (1.94).

¹Estimations based solely on graduates provides qualitatively similar results as those on the whole sample.

Table	e 1:	Va	riables	descr	iption
-------	------	----	---------	-------	--------

	Ν	Mean	SD	Min	Max
DEPENDENT VARIABLE					
Dig. affects my job satisfaction	3089	3.47	0.91	1	5
MAIN EXPLANATORY VARIABLES					
Dig. increases the time pressure at work	3089	3.23	1.18	1	5
Dig. puts my job at risk	3089	1.94	1.06	1	5
Dig. worsens the work-life balance	3089	2.72	1.18	1	5
Dig. leads to a smooth transition between working hours and leisure time	3089	3.01	1.2	1	5
Dig. makes my work more interesting	3089	3.4	1.09	1	5
Dig. reduces the proportion of repetitive tasks	3089	3.24	1.16	1	5
Dig. increases my productivity	3089	3.65	1.04	1	5
Dig. increases my autonomy at work	3089	3.15	1.06	1	5
Dig. enables more flexible forms of working time	3089	3.25	1.35	1	5
Dig. simplifies interactions with colleagues and superiors	3089	3.41	1.11	1	5
CONTROL VARIABLES					
How strongly does Dig affect the work over the last year?	3089	3 44	1 1 4	1	5
Women	3089	0.19	0.4	Ô	1
Age	3089	35.7	9.65	20	72
Executive (dummy for being firm's board director	3099	0.29	0.45	0	1
or member of management)				-	_
Field of study					
Agronomy	3089	0.01	0.1	0	1
Catering	3089	0.05	0.22	0	1
Health	3089	0.06	0.25	0	1
Arts	3089	0.01	0.1	0	1
Social work and adult education	3089	0.04	0.19	0	1
Technology	3089	0.66	0.47	0	1
Business administration	3089	0.17	0.37	0	1
Industry					
Manufacturing	3089	0.33	0.47	0	1
Construction	3089	0.12	0.32	0	1
Wholesale and retail trade; repair of motor vehicles and motorcycles	3089	0.03	0.16	0	1
Transportation and storage	3089	0.03	0.17	0	1
Accommodation and food service activities	3089	0.04	0.2	0	1
Information and communication	3089	0.08	0.27	0	1
Financial and insurance activities	3089	0.05	0.21	0	1
Professional, scientific and technical activities	3089	0.06	0.25	0	1
Administrative and support service activities	3089	0.08	0.27	0	1
Public administration and defence; compulsory social security	3089	0.05	0.21	0	1
Education	3089	0.04	0.19	0	1
Human health and social work activities	3089	0.1	0.3	0	1
Other service activities	3089	0	0.06	0	1

If digitalization has only a moderate effect on a given job characteristic, we hardly identify the overall effect of digitalization on job satisfaction through this job characteristic, and thus independently on the effect that this job characteristic has on job satisfaction. Therefore, these results cast doubt on both hypotheses H3 and H5. However, these two low values do not necessarily mean that work-life-balance and the fear of losing one's job have no effect on job satisfaction. Instead, it means that digitalization does not affect them.

The control variables in the bottom part of Table 1 show that most respondents are male and between ages 20 and 72. The average age of respondents is about 36 years, meaning that our sample is relatively young. About 30% hold executive positions, either as a member of a firm's board of directors or as part of management. The summary statistics show that about two-thirds of respondents chose technology-related field of study. About one sixth are in business administration, while the remaining sixth are subdivided among the other five fields. Finally, for the industry of activity, Table 1 shows that one third of the respondents are active in manufacturing. Moreover, financial and insurance activities, as well as human health and social work activities, represent a large portion of the sample.

A comparison between these summary statistics and the values collected by the Swiss Federal Statistical Office (SFO) through the Survey on Professional Education² suggests that our sample is not completely representative of the specific subgroup of workers having a degree from a PET college. Concretely, our sample overrepresents men and graduates in technology-related fields. Nevertheless, the average age at graduation in our sample is in line with the ones reported by of respondents is close to the one reported by the SFO.

5 Estimation Results

5.1 Main Results

The first three columns of Table 2 show the estimation results of the reduced form model presented in equation 4. In column (4) we report the average effect of digitalization on job characteristics, while in column (5) we multiply it with the estimated coefficients. Finally, in column (6) we show the Shapley values, which describe the contribution of each regressor in the goodness-of-fit of the estimation in column (3).

The estimations in the first three columns differ in terms of control variables, e.g., column (1) contains no control variables. Overall, the 10 characteristics explain about 34.5% of the total variance in the effect of digitalization on job satisfaction. Column (2) controls for individual characteristics, and column (3) further controls for the influence of digitalization on work in the preceding year. We find that these control variables have hardly any influence on the estimated coefficients. While the additional control variables increase the percentage of explained variance, they do so only slightly, to 37.9%.

²https://www.bfs.admin.ch/bfs/en/home/statistics/education-science/diploma/ tertiary-advanced-professional-training.html

	CULUCITUDE TIL					
	(1)	(2)	(3)	(4)	(5)	(9)
	$\hat{\beta}_c$	$\hat{\beta}_c$	$\hat{\beta}_c$	$\widetilde{\theta}_c$	$\widehat{\beta}_c^* \widetilde{\theta}_c$	Percentage
						of Explained
						Variation
Dig. increases the time pressure at work	-0.0741^{***}	-0.0683***	-0.0776***	3.23	-0.21	1.7
	(0.014)	(0.014)	(0.0139)			
Dig. puts my job at risk	-0.0312^{**}	-0.0287**	-0.0343^{***}	1.94	-0.067	0.52
	(-0.0147)	(-0.015)	(-0.0148)			
Dig. worsens the work-life balance	-0.102^{***}	-0.0953***	-0.102^{***}	2.72	-0.278	3.45
	(-0.0151)	(-0.015)	(-0.0148)			
Dig. leads to a smooth transition between working hours and leisure time	-0.0056	0.00145	0.00174	3.01	0.5	0.35
	(-0.0139)	(-0.0139)	(-0.0137)			
Dig. makes my work more interesting	0.165^{***}	0.170^{***}	0.155^{***}	3.4	0.527	6.51
	(-0.0169)	(-0.0171)	(-0.0171)			
Dig. reduces the proportion of repetitive tasks	0.0138	0.0182	0.0193	3.24	0.063	1.22
	(-0.0137)	(-0.0137)	(-0.0136)			
Dig. increases my productivity	0.281^{***}	0.271^{***}	0.255^{***}	3.65	0.928	11.01
	(-0.0177)	(-0.0177)	(-0.0179)			
Dig. increases my autonomy at work	0.0405^{**}	0.0403^{**}	0.0380^{**}	3.15	0.119	1.94
	(-0.0167)	(-0.0167)	(-0.0164)			
Dig. enables more flexible forms of working time	0.0231^{**}	0.0344^{***}	0.0269^{**}	3.25	0.087	1.48
	(-0.0119)	(-0.0122)	(-0.0122)			
Dig. simplifies interactions with colleagues and superiors	0.0595^{***}	0.0541^{***}	0.0548^{***}	3.41	0.187	3.12
	(-0.0148)	(-0.0148)	(-0.0148)			
How strongly does Dig. affect the work over the last year?			0.112^{***}		0.386	3.24
			(-0.0134)			
Worker characteristics		\mathbf{Yes}	\mathbf{Yes}			3.37
N	3089	3089	3089			
R^2 (%)	34.5	36.2	37.9			
Notes: Columns (1) to (3) remorts the results of the OIS regression baring as dependent	he off offering	ffoct of digitalized	ation on job eat	iefaction	mhich is	maseinad on a

Table 9. Estimation results

five point Likert scale (1="less satisfied", 3="no change", 5="more satisfied"). Robust standard errors in parentheses. * p<0.10, ** p<0.05, *** p<0.01. Worker character teristics is a vector of control variables as described in Table 1 plus industry dummies according to 1-digit NACE Rev. 2 classification. Column (4) shows $\tilde{\theta}_c$, the average effect of digitalization on job characteristics as reported in Table 1. Column (5) multiplies the β_{OLS} from column (3) with the $\tilde{\theta_c}$ from column (4). Column (6) reports the Shapley values, which describe the absolute contribution of each regressor in the goodness-of-fit of the estimation in column (3). The coefficients of the OLS regression of column (3) test our hypotheses on the effect of each job characteristic on job satisfaction. The value for $\hat{\beta}_c \tilde{\theta}_c$ reported in column (5) and the Shapley values reported in column (6) allow us to quantify the importance of each channel.

We start by considering the channels of the dimension "time use". The OLS coefficient for time pressure at work is negative and statistically significant, suggesting that the increase in time pressure at work resulting from digitalization decreases job satisfaction. Column (4) reports the corresponding value of $\tilde{\theta}_c$, the impact of digitalization on time pressure, which is average. Thus, as column (5) shows, $\hat{\beta}_c \ \tilde{\theta}_c$ amounts to -0.25. Column (6) shows that the increase in time pressure due to digitalization accounts for about 1.7% of the total variance. This finding support hypothesis H1—that digitalization decreases job satisfaction through an increase in time pressure.

The second channel of the "time use" dimension is the fear of job loss. While OLS coefficients for the fear of losing one's job are also negative and statistically significant, they are lower than the coefficient for the increase in time pressure. Moreover, θ_c is relatively low. Thus the resulting value of $\beta_c \theta_c$ is particularly low. This channel explains about 0.5% of the overall variance. Nevertheless, we should not interpret this result as meaning that the fear of losing one's job has no effect on job satisfaction. Instead, in this case it means that digitalization has almost no effect on workers' job satisfaction in terms of that fear. Thus, while we confirm hypothesis H2—that digitalization decreases job satisfaction by increasing the fear of losing one's job—we find a relatively small effect magnitude for this channel in our sample. This finding, however, should be relativized given the relatively low unemployment probability of the sample consisting of workers with a degree from a PET college³.

The third channel of the "time use" dimension is work-life balance. The large and negative OLS coefficient suggests that this channel has the strongest negative effect on job satisfaction of all 10 channels. However, as with the previous channel, the relatively low value of θ_c reduces the value of $\beta_c \theta_c$. Given that this channel explains about 3.4% of the total variance, we find that the relatively high value of θ_c confirms hypothesis H3—that digitalization decreases job satisfaction by deteriorating the work-life balance.

The final channel in the "time use" dimension is the smoothness of transition between work and private life. The OLS coefficient for this channel is not statistically different from zero, and θ

³See the unemployment rate of workers with tertiary professional education https://www.bfs. admin.ch/bfs/de/home/statistiken/bildung-wissenschaft/bildungsindikatoren/themen/wirkung/ arbeitsmarktstatus.assetdetail.12527130.html

remains relatively low. As the resulting $\hat{\beta}_c \ \tilde{\theta}_c$ is also close to zero, this channel explains less than 0.4% of the total variance. Thus, a smoother transition between working hours and leisure due to digitalization does not affect job satisfaction. Our findings therefore do not support hypothesis H4—that digitalization decreases job satisfaction by smoothing the transition between working hours and leisure time.

For the dimension "new activities", the OLS coefficient for the interestingness of work is positive and statistically significant. Given the high value of $\tilde{\theta}_c$, the resulting $\hat{\beta}_c \tilde{\theta}_c$ is also high. This channel explains about 6.5% of the total variance. Our estimations thus support hypothesis H5—that the interestingness of work as a result of digitalization positively affects job satisfaction.

The second channel of the "new activities" dimension is percentage of repetitive tasks. The OLS coefficient for this channel is small and not statistically different from zero. Thus the resulting value of $\hat{\beta}_c \tilde{\theta}_c$ is low, even though $\tilde{\theta}_c$ is relatively high. This channel explains about 1.2% of the total variance. Nevertheless, the low value of $\hat{\beta}_c$ suggests that the reduction in the proportion of repetitive tasks as a result of digitalization does not markedly affect job satisfaction. Hypothesis H6—that digitalization increases job satisfaction by reducing the proportion of repetitive tasks—is thus not confirmed.

The third channel of the "new activities" dimension is productivity. The large and positive OLS coefficient suggests that this channel has the strongest effect on job satisfaction. Furthermore, this channel has the largest value of θ_c , meaning that digitalization affects workers' productivity particularly strongly. The combination of these two large values gives a very high value of $\hat{\beta}_c \theta_c$, showing the large contribution of this channel to explaining the effect of digitalization on job satisfaction. Indeed, this channel alone accounts for about 11% of the total variance. This result clearly supports hypothesis H7—that the increase in productivity caused by digitalization positively affects job satisfaction.

The fourth channel of the "new activities" dimension is autonomy. The OLS coefficient for the increase in autonomy is positive and statistically significant. However, the relatively low value of $\hat{\beta}_c$ multiplied by an average value of $\tilde{\theta}_c$ gives a relatively small value of $\hat{\beta}_c \tilde{\theta}_c$. This channel explains altogether about 1.9% of the total variance in the model, a finding suggesting that greater autonomy at work due to digitalization positively affects job satisfaction, and thus support hypothesis H8.

As for dimension "access to information", we observe that the OLS coefficient for the flexibility of working time is positive and statistically significant but relatively small. Given the





Notes: This figure shows the relative importance of the 10 channels in explaining the impact of digitalization on workers' job satisfaction. The width of the arrows represents the relative importance of the channel. Black stands for channels with negative effects on job satisfaction. Grey represents a positive effect. A dashed arrow indicates that digitalization via this channel has no statistically significant effect on job satisfaction.

average value of $\tilde{\theta}_c$, the resulting value of $\hat{\beta}_c \tilde{\theta}_c$ is relatively low. This indicator explains about 1.5% of the total variance. Therefore we find support for hypothesis H9, suggesting that more flexible forms of work stemming from digitalization increase job satisfaction.

For the coefficient of the dimension "communication tools", the OLS coefficient for the simplicity of interaction with colleagues and superiors is positive and statistically significant. Given the relatively high value of θ_c , the resulting $\hat{\beta}_c \theta_c$ is also relatively high. This channel explains about 3.1% of the overall variance, meaning that digitalization positively affects job satisfaction by simplifying interactions with colleagues and supervisors, and thus supports hypothesis H10. Figure 3 summarizes the main findings discussed thus far.

5.2 Heterogeneity Across Workers

By formulating the hypothesis in Section 2, we refer on the entire sample and refrain from refining them according to workers' individual characteristics. Nevertheless, the data offers information on workers' characteristics, which can be used to explore the heterogeneity of the results.

Figure 4 shows the effect of digitalization on the ten channels by subgroups of workers. In Sub-figure 4a we report the mean of women compared to men. This figure suggests that





digitalization affects job characteristics relatively less strongly for women. The only exceptions are the effects of digitalization on increasing the fear of job loss and the simplified interaction with colleagues or superiors. These two channels are equally affected across gender. This figure suggests thus an overall weaker impact of digitalization on women compared to men.

The situation is less clear-cut when subdividing the sample by age. Sub-figure 4b illustrates that digitalization has a relatively less strong effect on time pressure, work-life balance, and transition between work and private life for workers younger than 35 years. In contrast, digitalization increases the interestingness of work, productivity, and autonomy relatively stronger for young workers than for older ones.

Sub-figure 4c presents the comparison of workers having a managerial position compared to workers without managerial position. Digitalization increases the fear of job loss less strongly by management workers. There is no statistical difference regarding time pressure, interestingness of work, and autonomy. In contrast, digitalization has a stronger effect on workers with managerial position with regard to the worsening of the work-life balance, the smoothing of the transition between work and private life, as well as in term of reducing repetitive tasks and making work more flexible.

Finally, Sub-figure 4d compares the means of workers that graduated in a technology-related field—the one by far most diffused in our sample—compared to other workers. In this case, we observe that digitalization increases the fear of job loss relatively less strong for workers who studied in a technology-related field. In contrast, the effect of digitalization with regard to the increase in time pressure, the worsening of the work-life balance, smoothing transition between work and private life, the increase in the interestingness of work, the increase in autonomy as well as the increase in working time flexibility are relatively stronger affected compared to workers who have not studied in technology-related fields.

Similarly as in the previous subsection, we run the reduced form model described in equation 4 for sub samples of workers according to their individual characteristics. To ease the comparisons across subgroups we report in Figure 5 the Shapley values which allow us to quantify the relative importance of each channel. Sub-figure 5a reports the results according to respondents' gender; Sub-figure 5b according to their age; Sub-figure 5c according to their management position; and Sub-figure 5d according to their field of study. Tables A1–A4 in the Appendix reports the OLS estimates by workers' characteristics, which underpin the regressors' contribution to R^2 .

estimates by workers' Tables A1–A4 in the Appendix reports the OLS workers. Notes: This figure shows the percentage of vasriation explained by each channel across subgoups of characteristics, which underpin the regressors' contribution to R^2

Simplicity of interaction with colleagues/superiors

Morking time flexibility

Autonomy

Productivity

Percentage of repetivie tasks

Interestingness of work

Smoothness of transition of work and private life

Fear of losing one's job

Work-life balance

Time pressure

0

۱0

G

Technology

Technology

Non-







Figure 5: Contribution to \mathbb{R}^2 across sub-samples

Starting by looking at the heterogeneity across gender, Sub-figure 5a, shows that the channels of time pressure and particularly work-life balance are more harmful for women. In contrast, women profit more through the increase in interestingness of work and more autonomy. Finally, the productivity channel, which is the most relevant channel, is slightly less beneficial for women.

For the heterogeneity across age groups, Sub-figure 5b shows that the channel of losing one's job is slightly more detrimental for older workers. Furthermore, the deterioration of the worklife-balance is clearly more critical for older workers. In contrast, older workers profit more through the increase of productivity and autonomy. However, older workers benefit less from more interesting work. Similarly, the channel of working time flexibility is slightly less beneficial for older workers.

As for the heterogeneity across executive position, Sub-figure 5c shows that non-executive workers suffer slightly less from an increase in time pressure and the worsening of the work-lifebalance. Additionally, non-executive workers benefit more from the increase in productivity and from an easier interaction with colleagues and superiors. However, they profit less from more interesting work.

Finally, for the heterogeneity across field of study, Sub-figure 5d shows that the channel of losing one's job is clearly more harmful for workers outside technology-related fields. However, they suffer less for the worsening of the work-life balance. Furthermore, workers outside technology-related fields profit more through the increase in productivity. Nevertheless, workers outside technology-related fields benefit less from more autonomy and easier interaction with colleagues and superiors.

5.3 Robustness Check

Our main estimation controls for how strongly digitalization affected respondents' work over the last year, thereby accounting for unobserved characteristics related to both digitalization and satisfaction. Furthermore, this information can be used to address potential heterogeneity in the relevance and perception of different job characteristics. Concretely, we conduct a robustness check that interacts the overall effect of digitalization on work with the effect of digitalization on each job characteristic. To ease interpretation, we standardize the resulting interaction terms by assigning the same mean and variance of the original effect of digitalization on each job characteristic.

$\begin{array}{c c c c c c c c c c c c c c c c c c c $			
Dig. increases the time pressure at work 0.0776^{***} (0.0128) 0.13^{*} (0.0175)lig. increases the time pressure at work 0.0776^{***} (0.0128) 0.0776^{***} (0.0175)lig. puts my job at risk -0.0343^{***} (0.0133) -0.0251 (0.0137)lig. worsens the work-life balance -0.102^{***} (0.0137) -0.120^{***} (0.0171)lig. leads to a smooth transition between working hours and leisure time * How strongly Dig. affect the work over the last year? 0.00174 (0.0131)lig. makes my work more interesting 0.155^{***} (0.0171) 0.00758 (0.0171)lig. reduces the proportion of repetitive tasks 0.0193 (0.0128) 0.0267 (0.0178)lig. increases my productivity 0.255^{***} (0.0211) 0.0267 (0.0178)lig. increases my autonomy at work 0.0380^{***} (0.0259) 0.0269^{**} (0.0259)lig. enables more flexible forms of working time 0.0269^{***} (0.0141) 0.0365^{***} (0.0207)lig. enables more flexible forms of working time 0.0269^{***} (0.0126) 0.0772^{***} (0.0126)lig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0126) 0.0772^{***} (0.0126)lig. simplifie		(1)	(2)
Increases the time pressure at work $-0.07.6^{-1.4}$ (0.0128)* How strongly Dig. affect the work over the last year? 0.0343^{***} (0.0133)* How strongly Dig. affect the work over the last year? -0.0251 (0.0154)Dig. worsens the work-life balance -0.102^{***} (0.0137)* How strongly Dig. affect the work over the last year? -0.120^{***} (0.0131)* How strongly Dig. affect the work over the last year? -0.120^{***} (0.0131)* How strongly Dig. affect the work over the last year? 0.00174 (0.0131)* How strongly Dig. affect the work over the last year? 0.00758 (0.0171)Dig. makes my work more interesting 0.155^{***} (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0128)Dig. increases my productivity 0.255^{***} (0.0128)* How strongly Dig. affect the work over the last year? 0.0380^{***} (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0127)Dig. enables more flexible forms of working time 0.0269^{**} (0.0126)* How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0126)* How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0126)* How		0.0770***	010
* How strongly Dig. affect the work over the last year? 0.0128 Dig. puts my job at risk 0.0343^{****} (0.0133) * How strongly Dig. affect the work over the last year? 0.0251 (0.0137) * How strongly Dig. affect the work over the last year? 0.102^{****} (0.0131) * How strongly Dig. affect the work over the last year? 0.102^{****} (0.0131) * How strongly Dig. affect the work over the last year? 0.00758 (0.0171) Dig. leads to a smooth transition between working hours and leisure time * How strongly Dig. affect the work over the last year? 0.00758 (0.0171) Dig. makes my work more interesting 0.155^{****} (0.0154) * How strongly Dig. affect the work over the last year? 0.233^{***} (0.0212) Dig. reduces the proportion of repetitive tasks 0.0193 * How strongly Dig. affect the work over the last year? 0.225^{****} (0.0128) * How strongly Dig. affect the work over the last year? 0.225^{****} (0.0178) Dig. increases my productivity 0.255^{****} (0.0161) * How strongly Dig. affect the work over the last year? 0.0267 (0.0178) Dig. increases my autonomy at work 0.0265^{***} (0.0259) Dig. increases my autonomy at work 0.0380^{****} * How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.05658^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0546^{***} * How strongly Dig. affect the work over the last year? 0.0548^{***} * How strongly Dig. affect the work over the last year? 0.0548^{***} * How strongly Dig. affect the work over the last year? 0.0548^{***} * How strongly Dig. affect the work over the last year? 0.0136) * How strongly Dig. affect the work over the last year? 0.0126^{**} * How strongly Dig. affect the work over the last year? 0.0269^{**} * How strongly Dig. affect the work over the last year? 0.0126^{**} * How strongly Dig. affect the work over the last year? 0.0121 Worker characteristics Yeas Yeas Yeas Yeas Yeas Ye	Dig. increases the time pressure at work	-0.0770****	
* How strongly Dig. affect the work over the last year? Dig. puts my job at risk * How strongly Dig. affect the work over the last year? Dig. increases my autonomy at work * How strongly Dig. affect the work over the last year? Dig. enables more flexible forms of working time * How strongly Dig. affect the work over the last year? Dig. simplifies interaction with colleagues and superiors * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last		(0.0128)	a second dedu
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	* How strongly Dig. affect the work over the last year?		-0.107***
Dig. puts my job at risk -0.0343^{***} (0.0133)* How strongly Dig. affect the work over the last year? -0.0251 (0.0137)* How strongly Dig. affect the work over the last year? -0.102^{***} (0.0131)* How strongly Dig. affect the work over the last year? 0.00174 (0.0131)* How strongly Dig. affect the work over the last year? 0.00758 (0.0171)Dig. nakes my work more interesting 0.155^{***} (0.0171)Dig. reduces the proportion of repetitive tasks 0.0193 (0.0178)Pig. increases my productivity 0.255^{***} (0.0178)Dig. increases my autonomy at work 0.0380^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0259)Dig. enables more flexible forms of working time 0.0269^{**} (0.0207)We strongly Dig. affect the work over the last year? 0.0516^{***} (0.0259)Dig. increases my productivity 0.255^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0207)We strongly Dig. affect the work over the last year? 0.0516^{***} (0.0207)Dig. enables more flexible forms of working time * How strongly Dig. affect the work over the last year? 0.0516^{***} (0.0153)Dig. simplifies interaction with colleagues and superiors * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0126)* How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0126)* How strongly Dig. affect the work over the last year? 0.0268^{***} (0.0126)* How strongly Dig. affect the work over the last yea			(0.0175)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Dig. puts my job at risk	-0.0343***	
* How strongly Dig. affect the work over the last year? -0.0251 (0.0154) Dig. worsens the work-life balance -0.102^{***} (0.0137) * How strongly Dig. affect the work over the last year? -0.20^{***} (0.0171) Dig. leads to a smooth transition between working hours and leisure time 0.00174 (0.0131) * How strongly Dig. affect the work over the last year? 0.00758 (0.0171) Dig. makes my work more interesting 0.155^{***} (0.0154) * How strongly Dig. affect the work over the last year? 0.233^{***} (0.0231) Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128) * How strongly Dig. affect the work over the last year? 0.225^{***} (0.0178) Dig. increases my productivity 0.255^{***} (0.0161) * How strongly Dig. affect the work over the last year? 0.0267 (0.0259) Dig. increases my autonomy at work 0.380^{***} (0.0259) Dig. increases my autonomy at work 0.0380^{***} (0.0147) * How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0259) Dig. increases my autonomy at work 0.0380^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0116) * How strongly Dig. affect the work over the last year? 0.0260^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0121^{***} * 0.0261^{***} (0.0126) * How strongly Dig. affect the work over the last year? 0.0264^{***} (0.0126) * How strongly Dig. affect the work over the last year? 0.0122^{***} * 0.0260^{***} (0.0126) * $0.0261^$		(0.0133)	
Dig. worsens the work-life balance (0.0154) big. worsens the work-life balance 0.012^{***} * How strongly Dig. affect the work over the last year? 0.0174 Dig. leads to a smooth transition between working hours and leisure time 0.00174 * How strongly Dig. affect the work over the last year? 0.00758 Dig. makes my work more interesting 0.155^{***} * How strongly Dig. affect the work over the last year? 0.233^{***} (0.0128)** How strongly Dig. affect the work over the last year? 0.0267 (0.0178) 0.0267 Dig. increases my productivity 0.255^{***} Dig. increases my productivity 0.255^{***} 0.0267 (0.0178) Dig. increases my autonomy at work 0.380^{***} (0.0147) 0.0516^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} Dig. increases my autonomy at work 0.0380^{***} (0.0127) 0.0516^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} Dig. simplifies interaction with colleagues and superiors 0.0348^{**} (0.0136) (0.0136) * How strongly Dig. affect the work over the last year? 0.0772^{***} Dig. simplifies interaction with colleagues and superiors 0.0772^{***} (0.0126) (0.0126) (0.0126) * How strongly Dig. affect the work over the last year? $(0.0126)^{**}$ Dig. simplifies interaction with colleagues and superiors $(0.0126)^{**}$ (0.0126) $(0.0126)^{**$	* How strongly Dig. affect the work over the last year?		-0.0251
Dig.worsens the work-life balance -0.102^{***} (0.0137)* How strongly Dig. affect the work over the last year? -0.120^{***} (0.0171)Dig. leads to a smooth transition between working hours and leisure time 0.00174 (0.0131)* How strongly Dig. affect the work over the last year? 0.00758 (0.0171)Dig. makes my work more interesting 0.155^{***} (0.0171)Dig. makes my work more interesting 0.155^{***} (0.0171)Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128)* How strongly Dig. affect the work over the last year? 0.233^{***} (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0267)Vis. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0126)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0161)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0121)How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0126)* How strongly Dig. affect the work over the last year? 0.01616^{**} (0.0126)* How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0			(0.0154)
	Dig. worsens the work-life balance	-0.102***	
* How strongly Dig. affect the work over the last year? $\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.0137)	
Dig. leads to a smooth transition between working hours and leisure time (0.0171) Dig. leads to a smooth transition between working hours and leisure time 0.00174 * How strongly Dig. affect the work over the last year? 0.00758 Dig. makes my work more interesting 0.155^{***} (0.0154) 0.0053 * How strongly Dig. affect the work over the last year? 0.233^{***} Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128) 0.0267 * How strongly Dig. affect the work over the last year? 0.0267 (0.0171) 0.0255^{***} Dig. increases my productivity 0.255^{***} * How strongly Dig. affect the work over the last year? 0.416^{***} (0.0161) (0.0161) * How strongly Dig. affect the work over the last year? 0.0380^{***} (0.0207) (0.0147) * How strongly Dig. affect the work over the last year? 0.0516^{***} (0.0120) (0.0147) * How strongly Dig. affect the work over the last year? 0.0516^{***} (0.0161) (0.0161) * How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0136) (0.0136) * How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0136) (0.0136) * How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0121) (0.0201) How strongly Dig. affect the work over the last year? 0.0260^{***} (0.0121) (0.0201) * How strongly Dig. affect	* How strongly Dig. affect the work over the last year?		-0.120***
Dig.leads to a smooth transition between working hours and leisure time 0.00174 (0.0131)* How strongly Dig. affect the work over the last year? 0.00758 (0.0171)Dig. makes my work more interesting 0.155^{***} (0.0154)* How strongly Dig. affect the work over the last year? 0.233^{***} (0.0231)Dig. reduces the proportion of repetitive tasks 0.0178 (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0259)i.g. increases my productivity 0.255^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.330^{***} (0.0127)* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0207)Dig. enables more flexible forms of working time 0.0269^{**} (0.0160)* How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0133)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0121)How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0121)Worker characteristicsYesYesN30893089R20.3790.381			(0.0171)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Dig. leads to a smooth transition between working hours and leisure time	0.00174	· · ·
* How strongly Dig. affect the work over the last year? (0.0171) Dig. makes my work more interesting (0.0154) * How strongly Dig. affect the work over the last year? (0.0231) Dig. reduces the proportion of repetitive tasks (0.0128) * How strongly Dig. affect the work over the last year? (0.0128) * How strongly Dig. affect the work over the last year? (0.0178) Dig. increases my productivity 0.255^{***} (0.0178) Dig. increases my productivity 0.255^{***} (0.0161) * How strongly Dig. affect the work over the last year? (0.0161) * How strongly Dig. affect the work over the last year? (0.0147) * How strongly Dig. affect the work over the last year? $(0.0269)^{**}$ (0.0207) Dig. enables more flexible forms of working time 0.0269^{**} (0.0116) * How strongly Dig. affect the work over the last year? (0.0163) Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136) * How strongly Dig. affect the work over the last year? (0.0136) * How strongly Dig. affect the work over the last year? (0.0136) * How strongly Dig. affect the work over the last year? (0.0136) * How strongly Dig. affect the work over the last year? (0.0136) * How strongly Dig. affect the work over the last year? (0.0136) * How strongly Dig. affect the work over the last year? (0.0121) (0.0121) $(0.0281)Worker characteristics Yes YesNNNNN$	0	(0.0131)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	* How strongly Dig. affect the work over the last year?	()	0.00758
Dig.makes my work more interesting 0.155^{***} (0.0154)* How strongly Dig. affect the work over the last year? 0.233^{***} (0.0231)Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.380^{***} (0.0147)* How strongly Dig. affect the work over the last year? 0.0267^{**} (0.0259)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0123)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0126)How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0126)Worker characteristicsYes YesN N R^2 3089 3089			(0.0171)
Dig. makes by torm information motion of the sector of	Dig makes my work more interesting	0 155***	(0.0 = 1 =)
* How strongly Dig. affect the work over the last year? (0.0231) Dig. reduces the proportion of repetitive tasks * How strongly Dig. affect the work over the last year? Dig. increases my productivity * How strongly Dig. affect the work over the last year? Dig. increases my autonomy at work * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? * How strongly Dig. affect the work over the last year? Dig. simplifies interaction with colleagues and superiors * How strongly Dig. affect the work over the last year? Dig. simplifies interaction with colleagues and superiors * How strongly Dig. affect the work over the last year? How strongly Dig. affect the work over the last year? How strongly Dig. affect the work over the last year? Morker characteristics N N R ² N N N N N N N N	2.6 manos mj tom more mereseng	(0.0154)	
How strongly Dig. Infect the work over the last year. (0.0231) Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128) * How strongly Dig. affect the work over the last year? 0.0267 (0.0178) Dig. increases my productivity 0.255^{***} (0.0161) * How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259) Dig. increases my autonomy at work 0.0380^{***} (0.0147) * How strongly Dig. affect the work over the last year? 0.0360^{***} (0.0207) Dig. enables more flexible forms of working time 0.0269^{**} (0.0161) * How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153) Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196) How strongly Dig. affect the work over the last year? 0.0116 * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196) How strongly Dig. affect the work over the last year? 0.012^{***} (0.0121) (0.0281) Worker characteristicsYes YesN 3089 3089 3089 3089	* How strongly Dig affect the work over the last year?	(0.0101)	0.233***
Dig. reduces the proportion of repetitive tasks 0.0193 (0.0128)* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0161)* How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0147)* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.027)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0121)Worker characteristicsYes YesN 3089 3089 R ² 3089 0.379	now sublight Dig. ancet the work over the last year.		(0.0231)
Dig. reduces the proportion of repetrive tasks (0.0133) * How strongly Dig. affect the work over the last year? (0.0128) Dig. increases my productivity 0.255^{***} * How strongly Dig. affect the work over the last year? 0.416^{***} (0.0125) 0.416^{***} (0.0259) 0.0380^{***} Dig. increases my autonomy at work 0.0380^{***} * How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0147) (0.0147) * How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0116) (0.0116) * How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0136) (0.0136) * How strongly Dig. affect the work over the last year? 0.0548^{***} (0.0136) (0.0136) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136) (0.0121) (0.0281) Worker characteristicsYesYesN 3089 3089 R^2 0.379 0.381	Dig reduces the properties of repetitive tasks	0.0103	(0.0251)
* How strongly Dig. affect the work over the last year? 0.0267 (0.0178)Dig. increases my productivity 0.255^{***} (0.0161) (0.0178) * How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0147)* How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0207)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0113)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0121)How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0128)Worker characteristicsYesYesN30893089 3089	Dig. reduces the proportion of repetitive tasks	(0.0133)	
How strongly Dig. affect the work over the last year? (0.0267) (0.0178) Dig. increases my productivity 0.255^{***} (0.0161) * How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259) Dig. increases my autonomy at work 0.0380^{***} (0.0147) * How strongly Dig. affect the work over the last year? 0.0269^{**} (0.0207) Dig. enables more flexible forms of working time 0.0269^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153) Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196) How strongly Dig. affect the work over the last year? 0.0122^{**} (0.0121) Worker characteristicsYes YesN R^2 3089 0.379 N R^2 3089 0.379	* How strongly Dig effect the work even the last year?	(0.0128)	0.0267
Dig. increases my productivity (0.0178) (0.0161) * How strongly Dig. affect the work over the last year? 0.255^{***} (0.0161) * How strongly Dig. affect the work over the last year? 0.0380^{***} (0.0147) * How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0207) Dig. enables more flexible forms of working time 0.0269^{**} (0.0116) * How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153) Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136) * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136) How strongly Dig. affect the work over the last year? 0.0121 (0.0121) How strongly Dig. affect the work over the last year? 0.112^{***} (0.0121) Worker characteristicsYes YesN R^2 3089 0.379 N R^2 3089 0.379	How strongly Dig. anect the work over the last year:		(0.0207)
Dig. increases my productivity 0.253^{+++} * How strongly Dig. affect the work over the last year? 0.0161)* How strongly Dig. affect the work over the last year? 0.0380^{***} * How strongly Dig. affect the work over the last year? 0.0269^{**} Dig. enables more flexible forms of working time 0.0269^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} Dig. simplifies interaction with colleagues and superiors 0.0548^{***} * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136) (0.0196) * How strongly Dig. affect the work over the last year? 0.112^{***} 0.0269^{**} (0.0121) (0.0121) (0.0281) Worker characteristicsYesN 3089 3089 3089 R^2 0.379 0.379 0.381		0.055***	(0.0178)
	Dig. increases my productivity	0.255^{+++}	
* How strongly Dig. affect the work over the last year? 0.416^{***} (0.0259)Dig. increases my autonomy at work 0.0380^{***} (0.0147)* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0207)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196)How strongly Dig. affect the work over the last year? 0.0112^{***} (0.0121)Worker characteristicsYesN R^2 3089 0.379 N R^2 0.379 Output 0.381		(0.0161)	0 11 0 * * *
Dig. increases my autonomy at work 0.0380^{***} * How strongly Dig. affect the work over the last year? 0.0516^{**} * How strongly Dig. affect the work over the last year? 0.0269^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} * How strongly Dig. affect the work over the last year? 0.0365^{**} Dig. simplifies interaction with colleagues and superiors 0.0548^{***} * How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0136) (0.0196) * How strongly Dig. affect the work over the last year? 0.112^{***} (0.0196) 0.0281 Worker characteristics Yes N 3089 3089 R^2 0.379 0.381	* How strongly Dig. affect the work over the last year?		0.416^{***}
Dig. increases my autonomy at work 0.0380^{***} (0.0147)* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0207)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196)How strongly Dig. affect the work over the last year? 0.112^{***} (0.0121)Worker characteristicsYesN3089 R^2 0.379 0.381			(0.0259)
$\begin{array}{cccc} (0.0147) \\ & \mbox{ How strongly Dig. affect the work over the last year?} & 0.0516^{**} \\ & (0.0207) \\ & \mbox{ Dig. enables more flexible forms of working time} & 0.0269^{**} \\ & (0.0116) \\ & \mbox{ How strongly Dig. affect the work over the last year?} & 0.0365^{**} \\ & (0.0153) \\ & \mbox{ Dig. simplifies interaction with colleagues and superiors} & 0.0548^{***} \\ & (0.0136) \\ & \mbox{ How strongly Dig. affect the work over the last year?} & 0.0772^{***} \\ & (0.0196) \\ & \mbox{ How strongly Dig. affect the work over the last year?} & 0.112^{***} & -0.260^{***} \\ & (0.0196) \\ & \mbox{ How strongly Dig. affect the work over the last year?} & 0.112^{***} & -0.260^{***} \\ & (0.0121) & (0.0281) \\ & \mbox{ Worker characteristics} & & Yes & Yes \\ \hline N & 3089 & 3089 \\ & R^2 & 0.379 & 0.381 \\ \end{array}$	Dig. increases my autonomy at work	0.0380***	
* How strongly Dig. affect the work over the last year? 0.0516^{**} (0.0207)Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196)How strongly Dig. affect the work over the last year? 0.112^{***} (0.0121)Worker characteristicsYesN 3089 3089 R^2 0.379 0.379 0.381		(0.0147)	
$\begin{array}{cccc} & & & & & & & & & & & & & & & & & $	* How strongly Dig. affect the work over the last year?		0.0516**
Dig. enables more flexible forms of working time 0.0269^{**} (0.0116)* How strongly Dig. affect the work over the last year? 0.0365^{**} (0.0153)Dig. simplifies interaction with colleagues and superiors 0.0548^{***} (0.0136)* How strongly Dig. affect the work over the last year? 0.0772^{***} (0.0196)How strongly Dig. affect the work over the last year? 0.112^{***} (0.0121)Worker characteristicsYesN R^2 3089 0.379 0.379 0.381			(0.0207)
	Dig. enables more flexible forms of working time	0.0269**	
* How strongly Dig. affect the work over the last year? Dig. simplifies interaction with colleagues and superiors * How strongly Dig. affect the work over the last year? How strongly Dig. affect the work over the last year? How strongly Dig. affect the work over the last year? Morker characteristics N R ² 0.0365^{**} (0.0153) 0.0772^{***} (0.0196) $0.0121**$ (0.0121) (0.0281) Ves Ves 3089 3089 0.379 0.381		(0.0116)	
$\begin{array}{c} (0.0153) \\ (0.0153) \\ \hline \\ \text{Dig. simplifies interaction with colleagues and superiors} \\ & & & & & & & & & & & & & & & & & & $	* How strongly Dig. affect the work over the last year?		0.0365**
$\begin{array}{cccc} \text{Dig. simplifies interaction with colleagues and superiors} & 0.0548^{***} \\ & (0.0136) \\ & & \\ &$			(0.0153)
$\begin{array}{c} (0.0136) \\ * \mbox{ How strongly Dig. affect the work over the last year?} & 0.0772^{***} \\ & (0.0196) \\ & (0.0196) \\ & (0.0121) & (0.0281) \\ \hline & Worker characteristics & Yes & Yes \\ \hline N & 3089 & 3089 \\ R^2 & 0.379 & 0.381 \\ \end{array}$	Dig. simplifies interaction with colleagues and superiors	0.0548***	
$\begin{array}{c} {}^{*} \mbox{ How strongly Dig. affect the work over the last year?} & 0.0772^{***} \\ & (0.0196) \\ {}^{How strongly Dig. affect the work over the last year?} & 0.112^{***} & -0.260^{***} \\ & (0.0121) & (0.0281) \\ \\ \hline Worker characteristics & Yes & Yes \\ \hline N & 3089 & 3089 \\ R^2 & 0.379 & 0.381 \\ \end{array}$		(0.0136)	
How strongly Dig. affect the work over the last year? (0.0196) $-0.260***$ (0.0121) Worker characteristicsYesN3089 R^2 0.3790.381	* How strongly Dig. affect the work over the last year?		0.0772 ***
How strongly Dig. affect the work over the last year? 0.112^{***} -0.260^{***} (0.0121) (0.0281) Worker characteristics Yes Yes N 3089 3089 R^2 0.379 0.381			(0.0196)
$\begin{array}{c c} (0.0121) & (0.0281) \\ \hline Worker characteristics & Yes & Yes \\ \hline N & 3089 & 3089 \\ R^2 & 0.379 & 0.381 \\ \end{array}$	How strongly Dig. affect the work over the last year?	0.112***	-0.260***
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		(0.0121)	(0.0281)
$egin{array}{cccc} N & 3089 & 3089 \ R^2 & 0.379 & 0.381 \end{array}$	Worker characteristics	Yes	Yes
R^2 0.379 0.381	N	3089	3089
	R^2	0.379	0.381

Table 9. Tobabilion Cheek	Table	3:	Robustness	Check
---------------------------	-------	----	------------	-------

Notes: This table reports the results of the OLS regression having as dependent variable the effect of digitalization on job satisfaction, which is measured on a five point Likert scale (1="less satisfied", 3="no change", 5="more satisfied"). Robust standard errors in parentheses.* p<0.10, ** p<0.05, *** p<0.01. Worker characteristics is a vector of control variables as described in Table 1 plus industry dummies according to 1-digit NACE Rev. 2 classification. Column (1) reports the baseline estimation as in Table 2, while column (2) reports the estimation of the model in which we allow the channel to interact with the overall effect of digitalization on work.

Table 3 shows the results of this robustness test. Column (1) reports the baseline estimation as in Table 2, while column (2) reports the estimation of the model in which we allow the channel to interact with the overall effect of digitalization on work. The comparison of the coefficients of these two columns suggests that indeed digitalization might partially strengthen or weaken the effect of the channels. Nevertheless, the signs and the sizes of the coefficients are similar, supporting thus our baseline results.

6 Conclusion, limitations, and implications

Using graduates of PET colleges in Switzerland as a case study, this paper provides insights into the relative strength of 10 channels through which digitalization affects job satisfaction. We find that digitalization increases job satisfaction among PET graduates particularly by increasing work productivity, making work more interesting, and fostering interactions with coworkers and supervisors. Relatively less important is the positive effect of digitalization on job satisfaction through the increase in workers' autonomy and more flexible forms of work.

Our results further suggest a negative effect of digitalization on job satisfaction through the worsening of the work life balance and an increase in time pressure. While the widespread idea that the fear of losing one's job to digitalization negatively affects job satisfaction is confirmed, it remains small in magnitude in our sample. Finally, our estimates provide no evidence that (a) digitalization negatively affects job satisfaction by smoothing the transition between work and private life, or (b) digitalization positively affects job satisfaction by reducing the proportion of repetitive tasks.

We further investigate the heterogeneity of our results by decomposing the sample according to respondents' gender, age, management position, and field of study. By comparing the relative contribution of the channels, we find relatively similar patterns across sub-samples. Major differences occur only for the effect that digitalization has on job satisfaction through the worsening of the work-life balance, a finding more relevant for men, for workers aged more than 35 years (roughly the average age in our sample), for workers with an executive position, and for workers whose field of study technology-related. For the effect that digitalization has on job satisfaction through an increase in the interestingness of work, we find a larger effect for males and for workers younger than 35. In contrast, the effect that digitalization has on job satisfaction through an increase in autonomy is lower for young workers, for women, and for workers who did not study in technology-related fields. n terms of productivity, we find that digitalization is more beneficial for women, for older workers, for workers without an executive position, and for workers who did not study in technology-related fields. Finally, the positive effect that digitalization has on job satisfaction by simplifying interactions with colleagues and superiors is larger for non-executive workers than for executives.

One limitation in our study is that even though controlling for gender, age, field of study, industry, management position, and a measure of how strongly respondents assess the impact of digitalization on their job, barely affects our estimates, we cannot rule out the possibility that other work characteristics (e.g., occupation) might affect our results. Therefore, future research should investigate other work characteristics that might prove to be channels for the effect of digitalization on job satisfaction.

Another limitation arises from the fact that the sample considered in this study is not perfectly representative, especially with regard to gender and field of study. Nevertheless, the fact that the results between sub-samples are particularly robust partially reassures the strength of the results.

Additionally, our results need cautious interpretation because they are specific to the particular sample investigated in this paper, which mainly consists of workers with management-level positions. The findings may differ substantially for workers with no tertiary vocational education. For example, a recent investigation conducted by Pfrombeck et al. (2020) on a representative sample of Swiss workers shows that, on average, a high degree of digitalization in the immediate work environment has a negative effect on job satisfaction. Future research should therefore evaluate the extent to which our results hold for different types of workers (e.g. workers with no tertiary vocational education).

Finally, a last limitation lies in this paper's reliance on respondent self-assessments of the influence of digitalization on job satisfaction and various work characteristics. Nonetheless, the estimates are robust to our controlling for various individual characteristics and for the self-assessed impact of digitalization on work. Furthermore, our estimates provide insights into the relative strength of various channels through which digitalization affects job satisfaction. However, due to the empirical strategy of a reduced form estimation, we are unable to interpret the results in absolute terms. Therefore, future research should use measures of digitalization, work characteristics, and job satisfaction that allow the estimating of structural models that capture these concepts directly and are less prone to potential measurement error.

References

- Acemoglu, D. & Autor, D. (2011). Skills, tasks and technologies: Implications for employment and earnings. In *Handbook of labor economics*, volume 4 (pp. 1043–1171). Elsevier.
- Acemoglu, D. & Restrepo, P. (2018). Artificial intelligence, automation and work. Technical report, National Bureau of Economic Research.
- Agypt, B. & Rubin, B. A. (2012). Time in the new economy: The impact of the interaction of individual and structural temporalities on job satisfaction. *Journal of Management Studies*, 49(2), 403-428.
- Amichai-Hamburger, Y. & Hayat, Z. (2011). The impact of the internet on the social lives of users: A representative sample from 13 countries. Computers in Human Behavior, 27(1), 585–589.
- Askenazy, P. & Caroli, E. (2010). Innovative work practices, information technologies, and working conditions: Evidence for france. *Industrial Relations: A Journal of Economy and Society*, 49(4), 544-565.
- Autor, D. (2014). *Polanyi's paradox and the shape of employment growth*, volume 20485. National Bureau of Economic Research Cambridge, MA.
- Autor, D. (2015). Why are there still so many jobs? the history and future of workplace automation. Journal of economic perspectives, 29(3), 3-30.
- Ayyagari, R., Grover, V., & Purvis, R. (2011). Technostress: technological antecedents and implications. MIS quarterly, 831–858.
- Bloom, N., Garicano, L., Sadun, R., & Van Reenen, J. (2014). The distinct effects of information technology and communication technology on firm organization. *Management Science*, 60(12), 2859– 2885.
- Boswell, W. R. & Olson-Buchanan, J. B. (2007). The use of communication technologies after hours: The role of work attitudes and work-life conflict. *Journal of Management*, 33(4), 592–610.
- Brooks, S. (2015). Does personal social media usage affect efficiency and well-being? Computers in Human Behavior, 46, 26–37.
- Brynjolfsson, E. & McAfee, A. (2011). Race against the machine: How the digital revolution is accelerating innovation, driving productivity, and irreversibly transforming employment and the economy. Brynjolfsson and McAfee.
- Carlsson, B. (2004). The digital economy: what is new and what is not? Structural change and economic dynamics, 15(3), 245-264.
- Caselli, F. & Coleman, W. J. (2001). Cross-country technology diffusion: The case of computers. American Economic Review, 91(2), 328–335.
- Castellacci, F. & Tveito, V. (2018). Internet use and well-being: A survey and a theoretical framework. Research policy, 47(1), 308-325.
- Castellacci, F. & Viñas-Bardolet, C. (2019). Internet use and job satisfaction. Computers in Human Behavior, 90, 141–152.
- D'Addio, A. C., Eriksson, T., & Frijters, P. (2007). An analysis of the determinants of job satisfaction when individuals' baseline satisfaction levels may differ. *Applied economics*, 39(19), 2413–2423.
- Day, A., Scott, N., & Kelloway, E. K. (2010). Information and communication technology: Implications for job stress and employee well-being. In New developments in theoretical and conceptual approaches to job stress. Emerald Group Publishing Limited.
- Duxbury, L., Towers, I., Higgins, C., & Thomas, J. A. (2007). From 9 to 5 to 24/7: How technology has redefined the workday. In *Information resources management: Global challenges* (pp. 305–332). IGI Global.

- Frey, C. B. & Osborne, M. A. (2017). The future of employment: How susceptible are jobs to computerisation? *Technological forecasting and social change*, 114, 254–280.
- Gallie, D. & Russell, H. (2009). Work-family conflict and working conditions in western europe. Social Indicators Research, 93(3), 445-467.
- Golden, T. D. & Veiga, J. F. (2005). The impact of extent of telecommuting on job satisfaction: Resolving inconsistent findings. Journal of management, 31(2), 301–318.
- Graetz, G. & Michaels, G. (2018). Robots at work. Review of Economics and Statistics, 100(5), 753-768.
- Haas, M. R. & Hansen, M. T. (2007). Different knowledge, different benefits: Toward a productivity perspective on knowledge sharing in organizations. *Strategic management journal*, 28(11), 1133–1153.
- Hartley, J., Jacobson, D., Klandermans, B., & Van Vuuren, T. (1990). Job insecurity: Coping with jobs at risk. Sage Publications Ltd.
- Hendriks, P. (1999). Why share knowledge? the influence of ict on the motivation for knowledge sharing. Knowledge and process management, 6(2), 91-100.
- Huettner, F., Sunder, M., et al. (2012). Axiomatic arguments for decomposing goodness of fit according to shapley and owen values. *Electronic Journal of Statistics*, 6, 1239–1250.
- Huysman, M. & Wulf, V. (2006). It to support knowledge sharing in communities, towards a social capital analysis. *Journal of information technology*, 21(1), 40–51.
- Kelliher, C. & Anderson, D. (2010). Doing more with less? flexible working practices and the intensification of work. *Human relations*, 63(1), 83–106.
- Koku, E., Nazer, N., & Wellman, B. (2001). Netting scholars: Online and offline. American Behavioral Scientist, 44 (10), 1752–1774.
- Konradt, U., Hertel, G., & Schmook, R. (2003). Quality of management by objectives, task-related stressors, and non-task-related stressors as predictors of stress and job satisfaction among teleworkers. *European Journal of Work and Organizational Psychology*, 12(1), 61-79.
- Kristensen, N. & Johansson, E. (2008). New evidence on cross-country differences in job satisfaction using anchoring vignettes. *Labour economics*, 15(1), 96–117.
- Legner, C., Eymann, T., Hess, T., Matt, C., Böhmann, T., Drews, P., Mädche, A., Urbach, N., & Ahlemann, F. (2017). Digitalization: opportunity and challenge for the business and information systems engineering community. Business & information systems engineering, 59(4), 301-308.
- Limbu, Y. B., Jayachandran, C., & Babin, B. J. (2014). Does information and communication technology improve job satisfaction? the moderating role of sales technology orientation. *Industrial Marketing* Management, 43(7), 1236-1245.
- Lopes, H., Lagoa, S., & Calapez, T. (2014). Work autonomy, work pressure, and job satisfaction: An analysis of european union countries. *The Economic and Labour Relations Review*, 25(2), 306–326.
- Martin, L. & Omrani, N. (2015). An assessment of trends in technology use, innovative work practices and employees' attitudes in europe. *Applied Economics*, 47(6), 623–638.
- Mazmanian, M., O. W. J. Y. J. (2013). The autonomy paradox: The implications of mobile email devices for knowledge professionals. *Organization science*.
- McMurtrey, M. E., Grover, V., Teng, J. T., & Lightner, N. J. (2002). Job satisfaction of information technology workers: The impact of career orientation and task automation in a case environment. *Journal of Management Information Systems*, 19(2), 273–302.
- Melamed, S., Ben-Avi, I., Luz, J., & Green, M. S. (1995). Objective and subjective work monotony: effects on job satisfaction, psychological distress, and absenteeism in blue-collar workers. *Journal of Applied Psychology*, 80(1), 29.

- Moqbel, M., Nevo, S., & Kock, N. (2013). Organizational members' use of social networking sites and job performance. *Information Technology & People*.
- Morris, M. G. & Venkatesh, V. (2010). Job characteristics and job satisfaction: Understanding the role of enterprise resource planning system implementation. *Mis Quarterly*, 143–161.
- Nam, T. (2014). Technology use and work-life balance. Applied Research in Quality of Life, 9(4), 1017–1040.
- Pfrombeck, J., Feierabend, A., Schärrer, L., Kornblum, A., Grote, G., & Staffelbach, B. (2020). Schweizer human relations-barometer 2020: Digitalisierung und generationen.
- Pincus, J. D. (1986). Communication satisfaction, job satisfaction, and job performance. Human communication research, 12 (3), 395-419.
- Popma, J. (2013). The janus face of the'new ways of work': Rise, risks and regulation of nomadic work.
- Ragu-Nathan, T., Tarafdar, M., Ragu-Nathan, B. S., & Tu, Q. (2008). The consequences of technostress for end users in organizations: Conceptual development and empirical validation. *Information systems* research, 19(4), 417–433.
- Raziq, A. & Maulabakhsh, R. (2015). Impact of working environment on job satisfaction. Procedia Economics and Finance, 23, 717-725.
- Reisel, W. D., Probst, T. M., Chia, S.-L., Maloles, C. M., & König, C. J. (2010). The effects of job insecurity on job satisfaction, organizational citizenship behavior, deviant behavior, and negative emotions of employees. *International Studies of Management & Organization*, 40(1), 74–91.
- Requena, F. (2003). Social capital, satisfaction and quality of life in the workplace. Social indicators research, 61(3), 331-360.
- Rotman, D. (2013). How technology is destroying jobs. Technology Review, 16(4), 28–35.
- Salanova, M., Cifre, E., & Martin, P. (2004). Information technology implementation styles and their relation with workers' subjective well-being. International Journal of Operations & Production Management.
- Scandura, T. A. & Lankau, M. J. (1997). Relationships of gender, family responsibility and flexible work hours to organizational commitment and job satisfaction. Journal of Organizational Behavior: The International Journal of Industrial, Occupational and Organizational Psychology and Behavior, 18(4), 377-391.
- Shapley, L. S. (1953). A value for n-person games. Contributions to the Theory of Games, 2(28), 307-317.
- Tarafdar, M., Tu, Q., Ragu-Nathan, B. S., & Ragu-Nathan, T. (2007). The impact of technostress on role stress and productivity. Journal of management information systems, 24 (1), 301-328.
- Tarafdar, M., Tu, Q., & Ragu-Nathan, T. (2010). Impact of technostress on end-user satisfaction and performance. Journal of management information systems, 27(3), 303–334.
- Warr, P. (2003). 20 well-being and the workplace. Well-being: Foundations of hedonic psychology, 392.
- Zhao, S. (2006). Do internet users have more social ties? a call for differentiated analyses of internet use. *Journal of Computer-Mediated Communication*, 11(3), 844–862.

Appendix

Table A1: Sub-sample	e results: E	sumai	dan by Jan	gender		Δ	Vomen	
	$\widehat{\beta}_{c}^{(1)}$	$\widetilde{\theta_c} (\widetilde{2})$	$\widehat{\beta}_{c}^{*} \widetilde{\theta}_{c}^{c}$	(4) Percentage	$\widehat{\beta}_{c}^{(5)}$	β_{OLS}	$\widehat{\beta}_c^{(7)}$	(8) Percentage
				of Explained Variation				of Explained Variation
Dig. increases the time pressure at work	-0.0766***	3.27	-0.25	1.77	-0.0756**	3.05	-0.23	1.38
Dig. puts my job at risk	(0.0159) - 0.026	1.93	-0.05	0.46	(0.0295) - 0.0511	2.01	-0.1	0.64
Dig. worsens the work-life balance	$(0.0167) - 0.103^{***}$	2.75	-0.28	3.6	(0.0322)-0.100***	2.58	-0.26	2.9
Dig. leads to a smooth transition between working hours and leisure time	(0.0168) - 0.000794	3.08	0	0.31	(0.0314) 0.00563	2.73	0.02	0.68
Dig. makes my work more interesting	(0.0155) 0.160^{***}	3.47	0.56	6.72	(0.0294) 0.134^{***}	3.09	0.41	5.72
Dig. reduces the proportion of repetitive tasks	(0.0192) 0.0212	3.28	0.07	1.29	(0.039) 0.00635	3.07	0.02	0.87
Dig. increases my productivity	$(0.0153) \\ 0.251^{***}$	3.69	0.93	10.76	$(0.0317) \\ 0.262^{***}$	3.45	0.9	11.76
Dig. increases my autonomy at work	(0.02) 0.0468^{**}	3.2	0.15	2.23	(0.0395) -0.0015	2.94	0	0.99
Dig. enables more flexible forms of working time	(0.0186) 0.0313^{**}	3.32	0.1	1.58	(0.0344) 0.0117	2.94	0.03	1.13
Dig. simplifies interactions with colleagues and superiors	(0.0135) 0.0546^{***}	3.42	0.19	3.04	(0.0277) 0.0670^{**}	3.38	0.23	3.48
How strongly does Dig. affect the work over the last year?	$(0.0168) \\ 0.108^{***}$			3.07	(0.0317) 0.113^{***}			3.59
	(-0.015)				(0.0302)			
Worker characteristics	Yes			3.16	Yes			7.61
$\frac{N}{R^2}$ (%)	1042 37.99				$2047 \\ 40.73$			
Notes: Columns (1) and (5) report the results of the OLS regression having	as depender	it varië	uble the	effect of digitaliz	zation on job	satisfac	tion, whi	ch is measured
on a Likert scale ($1=$ less satisfied', $3=$ no change', $5=$ more satisfied'). Ko cteristics is a vector of control variables as described in Table 1 plus industi	bust standar ry dummies :	d erroi accordi	s in par ng to N	entneses. * p<0 ACE 1-digit clas	.10, ** p<0.0 sification. Co	o, *** p lumns (<0.01. \ 2) and (vorker cnara- 3) report the
sub-sample average values of the job characteristics. Columns (3) and (7) r average values of the characteristics Columns (4) and (8) report the parcer	eports the m mage of evol	ultiplic ined v	ation of	the betas from by each represen	columns (1) a r in the estim	nd (4) [.] ation ir	with the column	sub-sample (1)

Table A2: Sub-sam	ple results:	Estim	ation b	y age				
	Yo	ung (<	= 35 ye	ars)		<) plo	35 year	
	(1)	(5)	(3)	(4)	$(\tilde{2})$	(9)	(2)	(8)
	\widehat{eta}_c	θ_c	$\widehat{\beta}_c^* \widehat{\theta}_c$	Percentage	\widehat{eta}_c	β_{OLS}	$\widehat{\beta}_c^* \widehat{\theta}_c$	Percentage
				of Explained Variation				of Explained Variation
Dig. increases the time pressure at work	-0.0768***	3.14	-0.25	1.37	-0.0790***	3.35	-0.26	1.97
Die mutemerisch at miele	(0.0224)	1 09	0.05	96.0	(0.018)	1 08	80 U	0.82
DIS. PURE IIIY JOD do LIER	-0.0411	1.32	-0.00	0.20	-0.0203) (0.0203)	1.30	-0.00	60. 0
Dig. worsens the work-life balance	-0.126^{***}	2.62	-0.23	2.44	-0.0869^{***}	2.84	-0.36	4.79
Dig. leads to a smooth transition between working hours and leisure time	(0.0242) -0.00697	2.95	0.04	0.43	(0.0189) 0.0125	3.09	-0.02	0.52
	(0.0209)				(0.0183)			
Dig. makes my work more interesting	0.145^{***}	3.5	0.57	6.31	0.162^{***}	3.27	0.47	6.31
	(0.0257)		1	,	(0.0235)			
Dig. reduces the proportion of repetitive tasks	0.0177	3.25	0.06	1.01	(0.0181)	3.22	0.06	1.58
Dia incresce my nuclivity	(0.0213) 0.981***	3 77	0.87	00	(0.01 <i>1</i> 9) 030***	3.18	0.08	19.03
DB. Increases in Fronteening	(0.0269)	-		0	(0.0239)	0F-0	0.00	00.71
Dig. increases my autonomy at work	0.0569^{**}	3.22	0.08	1.22	0.0243	3.06	0.17	2.83
	(0.0258)				(0.0215)			
Dig. enables more flexible forms of working time	0.00806	3.24	0.14	2.0	0.0420^{***}	3.26	0.03	1.14
Dis. simplifies interactions with colleagues and superiors	(0.0194) 0.0431^{**}	3.47	0.22	2.95	(0.0156) 0.0641^{***}	3.33	0.14	3.16
	(0.0212)				(0.0206)			
How strongly does Dig. affect the work over the last year?	0.0846^{***}			4.44	0.136^{***}			2.37
	(0.0201)				(0.0182)			
Worker characteristics	$\mathbf{Y}_{\mathbf{es}}$			2.11	Yes			2.55
Ν	1331				1758			
R^2 (%)	33.56				40.98			
Notes: Columns (1) and (5) report the results of the OLS regression having	g as depender	t varia	ble the d	effect of digitaliz	ation on job s	atisfact	ion, whic	h is measured
on a Likert scale $(1="less satisfied", 3="no change", 5="more satisfied")$. Ro	obust standar	d erron	s in pare	entheses. $* p < 0$.	10, ** p<0.05	, *** •d ***	<0.01. W	orker chara-
cteristics is a vector of control variables as described in Table 1 plus indust	ry dummies a	uccordii	$_{ m N}$ to $N_{ m A}$	ACE 1-digit class	sification. Colu	umns (2	2) and (6) report the
sub-sample average values of the job characteristics. Columns (3) and (7) 1	reports the m	ıltiplic	ation of	the betas from a	columns (1) an	ıd (4) w	rith the s	ub-sample
average values of the characteristics. Columns (4) and (8) report the perce	ntage of expli	ined v	arian ce l	by each regressor	r in the estima	tion in	column	1).

Table A3: Sub-sample resu	lts: Estima	tion b	y execu	tive position				
		Non-E	xecutive	0		Exc	ecutive	
	(1)	<u>وي (ت</u>	$\widetilde{\mathfrak{s}}_{\mathfrak{s},\widetilde{\mathfrak{o}}}^{\mathfrak{s},\widetilde{\mathfrak{o}}}(3)$	(4)	(5)	(0)	() () () () () () () () () () () () () ((8)
	ρ_c	θ_c	p_c θ_c	rercentage of Explained Variation	∂c	STOC	$\beta c^{-} \theta c$	rercentage of Explained Variation
Dig. increases the time pressure at work	-0.0846***	3.24	-0.27	1.6	-0.0567**	3.21	-0.18	1.97
Dig. puts my job at risk	(0.0164) - 0.0398^{**}	1.97	-0.08	0.57	(0.0271) - 0.0277	1.88	-0.05	0.42
Dig. worsens the work-life balance	$(0.0175) \\ -0.101^{***}$	2.67	-0.27	3.16	(0.028) -0.0956***	2.83	-0.27	3.89
Dig. leads to a smooth transition between working hours and leisure time	(0.0177) 0.0161	2.94	0.05	0.41	(0.0285) -0.0423	3.19	-0.13	0.59
Dig. makes my work more interesting	(0.016) 0.142^{***}	3.39	0.48	6.2	(0.0276) 0.192^{***}	3.44	0.66	7.12
Dig. reduces the proportion of repetitive tasks	(0.0198) 0.0330^{**}	3.19	0.11	1.36	(0.0345) -0.0106	3.36	-0.04	1.11
Dig. increases my productivity	(0.0158) 0.258^{***}	3.63	0.94	11.26	(0.0273) 0.253^{***}	3.69	0.93	10.49
Dig. increases my autonomy at work	(0.0206) 0.0316	3.14	0.1	1.79	(0.0358) 0.0431	3.16	0.14	2.18
Dig. enables more flexible forms of working time	(0.0197) 0.0206	3.2	0.07	1.56	(0.0294) 0.0529^{**}	3.35	0.18	1.48
Dig. simplifies interactions with colleagues and superiors	(0.0142) 0.0592^{***}	3.38	0.2	3.52	(0.0249) 0.0453	3.48	0.16	2.39
How strongly does Dig. affect the work over the last vear?	$(0.0175) \\ 0.123^{***}$			3.92	(0.0282) 0.0855^{***}			1.99
	(0.0154)				(0.028)			
Worker characteristics	Yes			3.23	Yes			5.07
	2193				896 2			
R^{2} (%)	38.58				38.71			
Notes: Columns (1) and (5) report the results of the OLS regression having on a Likert scale $(1="as satisfied", 3="no change", 5="more satisfied")$. Ro	g as depender bust standar	ıt varia d erron	ble the s in par	effect of digitalizent antheses. * p<0	zation on job s. $10, ** p<0.05$	atisfact , *** p	ion, whic <0.01. W	h is measured orker chara-
cteristics is a vector of control variables as described in Table 1 plus indust	ry dummies	accordii	$\log to N_{i}$	ACE 1-digit clas	sification. Colu	umns (2	2) and (6) report the
sub-sample average values of the job characteristics. Columns (3) and (7) r average values of the characteristics Columns (4) and (8) report the nerver	eports the m ntage of expl	ultiplic ained w	ation of arian <i>c</i> e]	the betas from w each regresso	columns (1) an r in the estima	ıd (4) w ation in	rith the s column	ub-sample 1)

ladie A4: Dud-Sample re	Sults: Esuit	Lation	by пец	1 OI STUAY		E		
		Non-Të	chnolog	×		T.ech	nology	
	(1)	(3)	(3)	(4)	(5)	(9)	(2)	(8)
	\widehat{eta}_c	$\widetilde{\theta}_{c}$	$\widehat{\beta}_{c}^{*}\widetilde{\theta}_{c}^{*}$	Percentage	\widehat{eta}_c	β_{OLS}	$\widehat{\beta}_{c}^{*}\widetilde{\theta}_{c}^{*}$	Percentage
				of Explained				of Explained
				Variation				Variation
Dig. increases the time pressure at work	-0.0904***	3.05	-0.28	1.83	-0.0685***	3.32	-0.23	1.59
	(0.0231)				(0.0172)			
Dig. puts my job at risk	-0.0772***	2.04	-0.16	1.4	-0.00704	1.89	-0.01	0.22
	(0.0249)				(0.0181)			
Dig. worsens the work-life balance	-0.0615^{**}	2.55	-0.16	1.96	-0.127^{***}	2.8	-0.36	4.44
	(0.0252)				(0.0183)			
Dig. leads to a smooth transition between working hours and leisure time	-0.00343 (0.0243)	2.9	-0.01	0.41	0.00411 (0.0168)	3.07	0.01	0.36
Dig. makes my work more interesting	0.151^{***}	3.25	0.49	6.49	0.154^{***}	3.48	0.54	6.61
5	(0.0305)				(0.0209)			
Dig. reduces the proportion of repetitive tasks	0.0321	3.26	0.1	1.36	0.0147	3.23	0.05	1.14
	(0.0257)				(0.0161)			
Dig. increases my productivity	0.278^{***}	3.63	1.01	12.38	0.237^{***}	3.65	0.87	10.14
-	(0.0313)			1 7 7	(0.0214)	č) 7 0	
Dig. increases my autonomy at work	0.0237	3.02	0.07	1.57	0.0476^{**}	3.21	0.15	2.23
	(0.0285)				(0.0199)			
Dig. enables more flexible forms of working time	0.0113	3.13	0.04	1.38	0.0321^{**}	3.31	0.11	1.58
· · · · · · · · · · · · · · · · · · ·	(0.0214)				(0.0145)			
Dig. simplifies interactions with colleagues and superiors	0.0327	3.44	0.11	2.68	0.0687***	3.4	0.23	3.38
- - - - - - - - - - - -	(0.0265)			((0.0174)			
How strongly does Dig. affect the work over the last year?	0.110^{***}			3.2	0.113^{***}			3.25
	(0.0237)				-0.016			
Worker characteristics	Yes			0.0442	Yes			3.3
Ν	1042				2047			
R^2 (%)	39.06				38.25			
Notes: Columns (1) and (5) report the results of the OLS regression having	g as dependen	t varial	ole the ϵ	effect of digitaliz	ation on job sa	utisfacti	on, whic	i is measured
on a Likert scale $(1="less satisfied", 3="no change", 5="more satisfied")$. Ro	bust standard	l errors	i in par€	ntheses. $* p<0$.	10, ** $p<0.05$,	>d ***	<0.01. W	orker chara-
cteristics is a vector of control variables as described in Table 1 plus indust	ry dummies a	ccordin	ig to N^{\prime}	ACE 1-digit class	sification. Colu	(2) mms (2)) and (6)	report the
sub-sample average values of the job characteristics. Columns (3) and (7) r	eports the m	ltiplica	tion of	the betas from c	columns (1) and	d (4) w	ith the s	ıb-sample
average values of the characteristics. Columns (4) and (8) report the percen	ntage of expla	ined va	riance b	y each regresso	r in the estimat	tion in	column (1).

fold of and. Ļ ito E 140 1 Table A4. Cub

ETH Zürich Chair of Education Systems Leonhardstrasse 21 8092 Zurich, Switzerland

www.ces.ethz.ch

Publisher: CES Chair of Education Systems Authors: Thomas Bolli & Filippo Pusterla Layout: ETH Zürich Photo: ETH Zürich

© ETH Zürich, March 2021